Author’s response to reviews

Title: Psychological health of women who have conceived using Assisted Reproductive Technology in Taiwan: Findings from a longitudinal study

Authors:

Mei-Zen Huang (meizen.huang@gmail.com)
Chien-Huei Kao (chenhuei@ntunhs.edu.tw)
Kuan-Chia Lin (kuanchia@ym.edu.tw)
Jiann-Loung Hwang (2401clinic@gmail.com)
Shuby Puthussery (shuby.puthussery@beds.ac.uk)
Meei-Ling Gau (meeiling@ntunhs.edu.tw)

Version: 2 Date: 05 Jul 2019

Author’s response to reviews:

Dear Reviewers,

Thank you very much for your kindness and your noteworthy comments and suggestions. They were very helpful for us when revising the paper. We appreciate the efforts you expended on this paper. We have carefully revised the manuscript according to your comments and suggestions. The following is the revision done by us in response.

Please represent authors' names using their initials, not their full name, in the Authors’ Contributions section of your Declarations. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Thanks for drawing our attention to this. The mistakes were corrected.

P28, Authors' contributions, line 1-3

Thank you for uploading your cover letter and response letter. As these files are no longer required at this stage of the editorial process, please remove them from your submission.

The cover letter has been removed.
Please upload your revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. Please also ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Thanks for drawing our attention to this. The revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. We ensure that all figures/tables and supplementary files are cited within the text.

Minor correction in Table 1. No need to give M+-Sd on the heading. All the characteristics are in percentages only. Therefore M+-SD can be deleted. provide percentages (%) and within a bracket number (n). Modify it.

The mistakes were corrected.

Table 1